Author's response to reviews

Title: Minimal percentage of the dose received by 90% of the urethra (%UD90) is the most significant predictor of PSA bounce in patients who underwent low-dose-rate brachytherapy (LDR-brachytherapy) for prostate cancer.

Authors:

Nobumichi Tanaka (sendo@naramed-u.ac.jp)
Isao Asakawa (iasakawa@naramed-u.ac.jp)
Kiyohide Fujimoto (kiyokun@naramed-u.ac.jp)
Satoshi Anai (jpjpg@hotmail.com)
Akihide Hirayama (hirayamn@naramed-u.ac.jp)
Masatoshi Hasegawa (hasegawa@naramed-u.ac.jp)
Noboru Konishi (nkonishi@naramed-u.ac.jp)
Yoshihiko Hirao (hiraoyos@naramed-u.ac.jp)

Version: 5 Date: 6 September 2012

Author's response to reviews: see over
Dear Dr Jigisha Patel, MRCP, PhD:

Enclosed please find our revised manuscript entitled “Minimal percentage of the dose received by 90% of the urethra (%UD90) is the most significant predictor of PSA bounce in patients who underwent seed implantation” by N. Tanaka et al., which we would like to submit for consideration for publication in “BMC Urology”.

We revised the original manuscript according to the reviewers’ comment as below.

Editor's Comments:
Needs major editing for spelling and grammar before we can accept.

The revised version had been corrected by professional native proofreader.

We all have consented to publication and the manuscript has not been nor will be published elsewhere. None of the authors have any conflicts of interest, including specific financial interests and relationships and affiliations relevant to the subjective matter or materials discussed in the manuscripts. No funding or financial support was received.

Thank you very much in advance for your kind consideration.

Sincerely yours,

Nobumichi Tanaka, M.D.
Department of Urology
Nara Medical University